Author’s response to reviews

Title: Incidence of Retinopathy of Prematurity Type 1 and Type 2 in a Regional Hospital of Social Security in the State of Queretaro, Mexico (2017-2018)

Authors:
Róger Iván Acevedo-Castellón (roger.acevedo@imoiap.edu.mx)
Paulina Ramírez-Neria (paulina.ramirez@imoiap.edu.mx)
Renata García-Franco (renata.garcia@imoiap.edu.mx)

Version: 6 Date: 26 Mar 2019

Author’s response to reviews:

March 26, 2019

Andreas Ebneter, M.D., Ph.D.
BMC Ophthalmology

Dear Dr. Ebneter:

I wish to again re-submit the manuscript titled “Incidence of Retinopathy of Prematurity Type 1 and Type 2 in a Regional Hospital of the Social Security in the State of Queretaro, Mexico (2017-2018)”. The manuscript ID is BOPH-D-18-00981R1.

I appreciate your thoughtful suggestions and insights and this opportunity improve the manuscript. I look forward to working with you and the reviewers to move this manuscript closer to publication in the BMC Ophthalmology.

The manuscript has been rechecked and the necessary changes have been made in accordance with your suggestions. The point-by-point responses to all comments have been prepared and provided below.

Sincerely,

Roger Acevedo-Castellón
Response to Comments

1. Please remove the file entitled "DATA.xlsx" from your manuscript as it contains identifying information. Currently, the exact dates of birth, in combination with other identifiers within your manuscript, may compromise patient/participant anonymity. We feel that this kind of data (birth dates) should not be disclosed outside of the hospital it is stored in.

   Thank you very much for your observations. We have done so.

2. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files

   Thank you very much for your observations. We have done so.